Author's response to reviews

Title: Symptom Burden in Chronic Kidney Disease; a population based cross sectional study

Authors:

Sameera Senanayake (sjsenanayake@yahoo.com)
Nalika Gunawardena (nalikaguna@hotmail.com)
Paba Palihawadana (ppalihawadana@unicef.org)
Palitha Bandara (palithabandara120@yahoo.com)
R Karunarathna (rhkarunarathna@gmail.com)
Priyantha Kumara (kumara151@gmail.com)
Rashan Haniffa (rashan@nicslk.com)

Version: 1 Date: 22 May 2017

Author’s response to reviews:

The authors' responses have been included in the covering letter